Author's response to reviews

Title: Incidence of and risk factors for Motor Neurone Disease in UK women: A prospective study

Authors:

Pat Doyle (pat.doyle@lshtm.ac.uk)
Anna Brown (anna.brown@ceu.ox.ac.uk)
Valerie Beral (pa.valerie.beral@ceu.ox.ac.uk)
Gillian Reeves (gillian.reeves@ceu.ox.ac.uk)
Jane Green (jane.green@ceu.ox.ac.uk)

Version: 2 Date: 28 December 2011

Author's response to reviews:

We have added details of ethical approval to the end of the first paragraph in the Methods section.